

## ANSWERING REVIEWERS

02/08/2013

Dear Editor,

**Title:** Chemoprevention of Hepatocellular Carcinoma in Patients with HCV-Related Cirrhosis

**Author:** Gianni Testino, Paolo Borro

**Manuscript No:** 3671

The manuscript has been improved according to the suggestions of reviewers:

1. Format has been updated
2. Revision has been made according to the suggestions of the reviewer
3. References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Hepatology*

Sincerely yours,

Dr. Borro Paolo

